Author’s response to reviews

Title: Low level of physical activity in women with rheumatoid arthritis is associated with cardiovascular risk factors but not with body fat mass - a cross sectional study

Authors:

Ann-Charlotte Elkan (ann-charlotte.elkan@karolinska.se)
Niclas Håkansson (niclas.hakansson@ki.se)
Johan Frostegård (johan.frostegard@ki.se)
Ingiäld Hafström (ingiald.hafstrom@karolinska.se)

Version: 3 Date: 12 January 2011

Author’s response to reviews: see over
Dear Editor,

The suggested changes have now been made.
1. All the highlighting and tracking have been removed.
2. Full citation details for references 4, 19, 25 and 33 have been provided. The reference list is also changed as recommended.
3. The table titles have been changed to one sentence each.

Sincerely,

Ann-Charlotte Elkan
RN, CNS, PhD
Rheumatology department, R92
Karolinska Institute at Karolinska University Hospital Huddinge
141 86 Stockholm
Sweden
Email: ann-charlotte.elkan@karolinska.se
Phone: +46 8 5858 4746; Fax: +46 8 5858 18 78